Author's response to reviews

Title: Heparin (GAG-hed) inhibits LCR activity of Human Papillomavirus type 18 by decreasing AP1 binding

Authors:

Rita Villanueva (rvr1032000@yahoo.com.mx)
Nestor Morales-Peza (nestor@correo.biomedicas.unam.mx)
Irma Castelan-Sanchez (irmacast@servidor.unam.mx)
Enrique Garcia-Villa (enrique.villa@excite.com)
Rocio Tapia (rociotp@fisio.cinvestav.mx)
Angel Cid-Arregui (cid@dkfz.de)
Alejandro Garcia-Carranca (carranca@correo.biomedicas.unam)
Esther Lopez-Bayghen (ebayghen@cinvestav.mx)
Patricio Gariglio (vidal@cinvestav.mx)

Version: 7 Date: 30 August 2006

Author's response to reviews: see over
BioMed Central Editorial Production Team

Regarding to:

BMC-series journals
MS: 1883308808103186

“Heparin (GAG-hed) inhibits LCR activity of Human Papillomavirus type 18 by decreasing AP1 binding”.

I am glad to inform you that all formatting changes requested were performed, as follow:
1. Superscript numbers to link the authors’ to their affiliations appear now.
2. Page 9 table/box – We included the text within the manuscript, up-dating references within the text.

We hope for all to be complete and satisfactory to your entire request.

Sincerely,

[Signature]

Professor Esther López-Bayghen, PhD